Author's response to reviews

Title: Double chambered right ventricle with severe calcification of tricuspid valve in an elderly woman: a case report

Authors:

Nozomu Tamai (tamain04mfk@peace.ocn.ne.jp)
Shigenori Ito (sito@higashi-hosp.jp)
Kotaro Morimoto (mrmtktr@hotmail.com)
Masahiko Inomata (inomata@gb3.so-net.ne.jp)
Takayuki Yoshida (tyoshida@higashi-hosp.jp)
Shin Suzuki (suzuki_shin@nifty.com)
Yoshimasa Murakami (yhyvilla@aol.com)
Koichi Sato (ksato@higashi-hosp.jp)

Version: 2 Date: 13 November 2010

Author's response to reviews:

Thank you very much for the decision of acceptance of our manuscript (Double chambered right ventricle with severe calcification of tricuspid valve in an elderly woman: a case report (Manuscript ID 1528212675417120)) for publication in Journal of Medical Case Reports.

We submit the revised version of the manuscript. The changed points are as following;

1. Patient ID: We corrected the patient’s information (to ‘a 71-year-old Japanese woman’) in the Abstract, as well as in the main text.

2. Figures: We removed all identifying text on the figures. (We removed dates and times from Figure 2)

3. Horizontal line: We removed the horizontal line underneath the figure legends section.

4. Typography: We corrected the typographical errors in the main text.

Sincerely,

Nozomu Tamai, MD
Division of Cardiology, East Medical Center, Higashi Municipal Hospital, City of Nagoya
E-mail: tamain04mfk@peace.ocn.ne.jp
Tel.: +81-52-721-7171